Author's response to reviews

Title: Punishment or harm reduction? Contraceptive access for women who use drugs

Authors:

Anna Olsen Dr (aolsen@kirby.unsw.edu.au)
Cathy banwell Dr (cathy.banwell@anu.edu.au)
Annie Madden (anniem@avil.org.au)

Version: 2 Date: 10 July 2013

Author's response to reviews:

The sentence below has been added to the first paragraph of the methods section.

Ethics approval was received from the Australian National University Human Research Ethics Committee and written consent was requested from all participants.